Author's response to reviews

Title: Features of Complementary and Alternative Medicine Use by Patients with Coronary Artery Disease in Beijing: A Cross-Sectional Study

Authors:

Fu-Yong Chu (fuyongchu@gmail.com)
Yan Xue (snowater27@hotmail.com)
Ze Zhang (13066796232@163.com)
Xing-Jiang Xiong (5administration@163.com)
Jie Wang (wangjie0103@yahoo.com.cn)
Hong-Xu Liu (lhx_@263.net)

Version: 5 Date: 9 October 2013

Author's response to reviews: see over
Dear Editor-in-Chief of Journal of BMC Complementary and Alternative Medicine,

We deeply appreciate the opportunity to make the modifications to the manuscript and to resubmit our article “Features of Complementary and Alternative Medicine Use by Patients with Coronary Artery Disease in Beijing: A Cross-Sectional Study”. We are sorry for our previous careless neglect of Review 1’s comments on Statistic Value Column of Table 2. During the previous revisions, we thought he meant that an explanation of the p value was needed.

The Statistic Values in Table 2 refer to chi-square and t values for each row; however, we would like to delete the whole column as Review 1 mentioned in his first round of comments to the article.

Thanks very much for all the help!

Best Regards!

Fu-Yong Chu MD, PhD
Department of Cardiology
Beijing Hospital of Traditional Chinese Medicine
Capital Medical University, Beijing, China
Cellphone: +86 15810499319,
Office Tell: 86-10-52176952
Email: fuyongchu@gmail.com.
Reply to Review 1

Dear Dr A Templeton,

Thanks very much for your meticulous reviewing of our manuscript, especially for the comments regarding the Statistic Value in Table 2 of the manuscript. We are sorry for our previous careless neglect of your comments on Statistic Value Column of Table 2. During the previous revisions, we thought you meant that an explanation of the p value was needed. Upon the reminding of Editors at the journal, we decided to accept your opinion and we deleted the column of Statistic Value in Table 2 of the manuscript. We deeply appreciate your help. Thanks very much!

Best Regards!

Fu-Yong Chu MD, PhD
Department of Cardiology
Beijing Hospital of Traditional Chinese Medicine
Capital Medical University, Beijing, China
Cellphone: +86 15810499319,
Office Tell: 86-10-52176952
Email: fuyongchu@gmail.com.